Author's response to reviews

Title: Partial anomalous pulmonary venous return and atrial septal defect in adult patients de-tected with 128-slice multidetector computed tomography

Authors:

Sari Kivistö (sari.kivisto@hus.fi)
Helena Hänninen (Helena.hanninen@hus.fi)
Miia Holmström (miia.holmstrom@hus.fi)

Version: 2 Date: 30 September 2011

Author's response to reviews: see over
September 30\textsuperscript{th} 2011

To:
Editors-in-chief
Mr Vipin Zamvar
Royal Infirmary of Edinburgh, United Kingdom

Prof David Taggart
John Radcliffe Hospital, United Kingdom

Section: Case Report

MS: Partially anomalous pulmonary venous return and atrial septum defect in adult patients detected with 128 - slice multidetector computed tomography

Please find enclosed the above referred revised manuscript, which we are re-submitting for publication in the \textit{Journal of Cardiothoracic Surgery}. We have responded to each of editor´s points and done recommended corrections.

Sincerely, Sari Kivistö, MD